



Article A Model-Aware Comprehensive Tool for Battery Energy Storage System Sizing

Matteo Spiller¹, Giuliano Rancilio¹, Filippo Bovera¹, Giacomo Gorni², Stefano Mandelli³, Federico Bresciani² and Marco Merlo^{1,*}

¹ Politecnico di Milano—Department of Energy, Via Lambruschini 4a, 20156 Milano, Italy; matteo.spiller@polimi.it (M.S.); giuliano.rancilio@polimi.it (G.R.); filippo.bovera@polimi.it (F.B.)

² Eni S.p.A., Renewable, New Energies and Material Science Research Center, Via Fauser 4, 28100 Novara, Italy; giacomo.gorni@eni.com (G.G.); federico.bresciani@eni.com (F.B.)

- ³ Plenitude, Via Giuseppe Ripamonti 85, 20141 Milano, Italy; stefano.mandelli@plenitude.com
- * Correspondence: marco.merlo@polimi.it

Abstract: This paper presents a parametric procedure to size a hybrid system consisting of renewable generation (wind turbines and photovoltaic panels) and Battery Energy Storage Systems (BESS). To cope with the increasing installation of grid-scale BESS, an innovative, fast and flexible procedure for evaluating an efficient size for this asset has been developed. The tool exploits a high-fidelity empirical model to assess stand-alone BESS or hybrid power plants under different service stacking configurations. The economic performance has been evaluated considering the revenue stacking that occurs when participating in up to four distinct energy markets and the degradation of the BESS performances due to both cycle- and calendar-aging. The parametric nature of the tool enables the investigation of a wide range of system parameters, including novel BESS control logic, market prices, and energy production. The presented outcomes detail the techno-economic performances of a hybrid system over a 20-year scenario, proposing a sensitivity analysis of both technical and economic parameters. The case study results highlight the necessity of steering BESS investment towards the coupling of RES and accurate planning of the service stacking. Indeed, the implementation of a storage system in an energy district improves the internal rate of return of the project by up to 10% in the best-case scenario. Moreover, accurate service stacking has shown a boost in revenues by up to 44% with the same degradation.

Keywords: battery energy storage system; renewables; market service stacking

1. Introduction

The energy sector is responsible for a large share of anthropogenic carbon emissions that lead to climate change and global warming. The 2015 United Nations Climate Change Conference in Paris set the milestone of limiting the average temperature increase to below 1.5 °C [1]. A total of 160 countries around the world have agreed to combat global heating through the installation of renewable energy sources (RES) for a more sustainable energy scenario. Consequently, the overall energy production by RES has increased from 20 to 28% since 2010, and the target is for production to reach at least 43% by 2030 [2]. Focusing on the European Union (EU), the EU Green Deal and the "Fit-for-55" package set the goal is of reducing greenhouse gas emissions by 55% with respect to 1990 by 2030 [3].

RES have a low carbon footprint and are, therefore, among the main candidates for energy sources that could be used to reach these decarbonization targets. In any case, their integration into the energy system is not straightforward. These sources are characterized by the high intermittency and non-programmability of the energy output, which complicates the balance between supply and demand in the power system [4]. Gridconnected battery energy storage systems (BESS) represent a viable resource to cope with those issues and guarantee the balance, stability, and adequacy of a decarbonizing power



Citation: Spiller, M.; Rancilio, G.; Bovera, F.; Gorni, G.; Mandelli, S.; Bresciani, F.; Merlo, M. A Model-Aware Comprehensive Tool for Battery Energy Storage System Sizing. *Energies* **2023**, *16*, 6546. https://doi.org/10.3390/en16186546

Academic Editor: Quanqing Yu

Received: 3 July 2023 Revised: 16 August 2023 Accepted: 25 August 2023 Published: 12 September 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). system. Compared to other storage systems, BESS are close to market maturity, with prices dropping by 87% from 2010 to 2019 [5]. The installed grid-scale battery storage capacity will expand 44-fold between 2021 and 2030 to 680 GW according to IEA [6]. In addition, lithium-ion batteries outperform other storage technologies in terms of energy density, power density, and round-trip efficiency. Moreover, their operational reliability can last up to 20 years with suitable management strategies, such as temperature regulation and capacity augmentation [7]. The success of this technology is attributed to the flexibility and scalability of these assets, coupled with their ability to behave as bulk energy systems. These characteristics enable the provision of a wide set of services to system operators and active users. Such services could reduce RES uncertainty and make BESS attractive investments.

Despite the possible enhancement that BESS could provide to electric power system operation, this technology still has shortcomings. Although battery prices are decreasing yearly, the overall cost per kWh is still high, representing the most common challenge. To cope with high capital costs, it is crucial to capture multiple cash flows to increase the financial viability of the project. The dynamic stacking of BESS services ensures higher profits for the asset [8]. Therefore, it is of paramount importance to evaluate an efficient system operation that creates multiple streams of revenues to increase the economic benefits. To properly assess and optimize the cash flow, the owner of the system must optimally size the BESS, accounting for several economic aspects such as cell technology, installation cost, and maintenance cost. These expenditures are mainly affected by lifetime, battery capacity, and overall performance, which interact in a non-linear manner. The proper evaluation of these aspects allows for more accurate modeling and, consequently, a more precise economic analysis.

High-fidelity BESS modelization is mandatory to ensure accurate economic evaluation. This paper proposes a model-aware BESS-sizing procedure that accurately represents the performance of BESS in different energy markets during their lifetime, accounting for the main non-linearities. In general, the stacking of the services is mostly addressed by constant BESS models that do not consider the non-linearities of this technology and the presence of auxiliaries. Furthermore, service stacking with grid-scale storage is mainly investigated in a stand-alone configuration, without evaluating the possible services that the system can provide to an RES power plant. Lastly, services such as the capacity market have not been assessed to date. Based on the state-of-art described in the next section, the novelties of this work are as follows:

- A sizing procedure is developed that investigates a 20-year BESS investment with a high-fidelity empirical model developed in [9] and updated with equations capable of emulating the capacity degradation of the system. The latter structure has been exploited to create a fast and flexible tool that is able to evaluate the most cost-effective storage investment, ensuring an efficient trade-off between computational effort and accuracy.
- Innovative algorithms are developed that are capable of stacking multiple services with a sequential approach. The implemented solutions produce results seamlessly, with two distinct configurations: stand-alone and hybrid-renewable power plants.

The work is structured as follows. Section 2 reviews the modelization and the algorithms exploited for sizing BESS in the literature. Section 3 describes the proposed empirical model, the methodology of the sizing procedure, and the novel algorithm proposed for stacking the energy markets. Section 4 introduces the study cases. Section 5 discusses the main results. Lastly, Section 6 summarizes the activities and lists future works.

2. Literature Review

The BESS sizing procedure consists of identifying the most cost-effective configuration for the stakeholders. The application is complex and non-linear. This section aims to describe two different fundamental aspects of the procedure: the modeling, and solution methods [10].

2.1. Modeling

Modeling focuses on the mathematical representation of the key components of BESS. A wide number of approaches have been developed, with different levels of complexity and computational effort. According to the degree of physical insight, battery models can be divided into three different levels: electrochemical model, circuit-oriented (or electrical) model, and black box models (empirical or stochastic) [11]. The selection of a model is bounded to the application area. Namely, a specific model ensures a different degree of accuracy and computational effort, and depending on the details required by the implementation, an efficient balance between the two characteristics improves the quality of the analysis. For instance, electrochemical models are the most accurate approach to battery representation. They describe the chemical reactions that take place in the electrodes and the electrolytes using a set of non-linear differential equations [12]. The expressions detail the effect of the electrochemical reactions, such as the diffusion, migration, kinetic phenomena, and lithium concentration, to truly represent the state variables of the system [13]. A description of the batteries at a microscopic scale accurately represents the key behaviors of the nonlinear system [14]. Although there is no doubt regarding the accuracy of electrochemical models, they are used in low-speed applications, such as online capacity estimation [15,16], predictive maintenance [17] or the validation of complementary models [18]. Circuit-oriented models are electrical equivalent models that are able to represent the state variables of the batteries [11]. Electrical models consist of electrical circuits made by capacitors and resistances, whose proper connection emulates the behavior of the batteries. The basic electrical model, known as Rint, has a big capacitor that can describe the open-circuit voltage of the cell, and a series resistance that can simulate the battery's internal resistance [19]. Moreover, RC circuits can be connected in series to the Rint model to increase the order of the system and emulate relaxation and polarization effects [20]. Despite the empirical nature of those elements, these modelizations are widely adopted thanks to their computational efficiency in capturing the dynamic response of the system [21]. The ability to estimate the state of batteries in real-time makes these models suitable for applications such as EV state estimation [20], and grid stability [22,23]. Furthermore, electrical models ensure a good performance when evaluating the terminal voltage and SOC of the batteries. Articulated modelization, such as a three-order model, allows for these state variables to be described with errors lower than 1% [21,24]. However, if the accuracy of the model increases, the same occurs regarding the computational effort [25]. In sizing applications, computational efficiency is a priority since it requires the evaluation of steady-state performances over a long time-window (e.g., 15–20 years). Therefore, the complexity of the electrical model could not produce results that are valuable for the procedure in a reasonable time. Empirical models ensure the optimal computational effort for this kind of analysis. The modelization is characterized by mathematically constructed models that utilize the observed data and measurements to represent the behavior and performance of BESS systems. These models are developed by an analysis of real-world operational data and the characteristics gathered from datasheets or experiments, allowing for them to capture the key relationships and patterns between various parameters [26,27]. By leveraging statistical techniques, regression analyses, or other mathematical approaches, empirical models provide a parametric representation of how different factors, such as battery lifetime, efficiency, and capacity, interact and impact the overall performance of BESS systems [9].

A fair share of BESS sizing procedures are based on scalar linear empirical systems that resemble the BESS performances. In [28], a constant battery and inverter efficiency model has been exploited to size a PV residential system by evaluating the economic criteria of the annuity method. Study [29] investigates the optimal sizing of BESS through a life-cycle cost model. This model inspects the different phases of the system, also accounting for the final decommissioning, maintenance and recycling, and disposal as a reduction in the economics of the system. Ref. [30] proposes a constant-efficiency empirical model to size an energy storage system, accounting for hosting capacity and reductions in wind

curtailment. Although these simplified empirical models represent the easiest approach to configuring and producing results for BESS, they are generally the least accurate [31]. Nevertheless, high-fidelity empirical models that represent varying efficiencies and power capabilities depending on SoC and C-rate improve the accuracy of the analysis [32]. These approaches rely on experimental campaigns to create a computationally efficient BESS model that contains an error. In [33], a regression technique has been exploited to build a non-linear BESS with an average SOC root mean square error of 3%. In [34], a detailed non-linear power losses model has been implemented to more precisely capture the low-efficiency working region of the storage system. In [9], a grid-scale BESS has been modeled with lookup tables to represent the non-linear efficiency and the auxiliary consumption of the system, ensuring an average SOC error of 0.168%. Despite the high fidelity of the modelization, the latter applications are limited to the operation perspective, and no one has used the accuracy of the model to size the BESS.

2.2. Solution Method

The solution method consists of the approach used to investigate the optimal size of the BESS, i.e., the mathematical procedure adopted to set the schedule and the dispatching of different services. The goal of the algorithm selected in the sizing procedure is to identify the best power setpoint to efficiently maximize the benefits generated by the BESS. The algorithm needs to emulate the provision of diverse and multiple services by the BESS. Namely, at each time step, given the actual SOC, the storage system is charged or discharged according to a logic that improves the welfare of the BESS owner. Most existing studies are based on mathematical programming due to their ability to identify a global optimum for the objective function under analysis. Stochastic Dynamic Programming (DP), and stochastic Mixed-Integer Linear Programming (MILP) or techniques derived from these two approaches are the most-adopted solutions to the sizing problem [35]. For instance, studies [36,37] propose a stochastic MILP and DP, respectively, to optimize the sizing of a grid-scale storage system with constant efficiency. However, these methodologies show important limitations in terms of the modeling and objectivity of the solution [38]. The optimal sizing problem is a non-convex and non-linear combinatorial optimization problem [39]. Therefore, MILP-sizing procedures need to rely on techniques such as relaxation, piecewise linear approximation, or the implementation of a constant efficiency to generate a problem that can be treated by the standard solvers [32,40,41]. However, DP is afflicted by dimensionality problems that need to be solved with the so-called approximated dynamic programming [42,43]. Although these techniques solve the main issues related to computational effort, their implementation comes at the expense of accuracy. Moreover, sizing problems are computationally intensive, and many works consider a timeframe that does not evaluate the whole investment lifespan. Lastly, to properly manage technical and economic targets, multi-objective functions are typically required [44]. These lead to a lack of objectivity in the solutions, which needs to be properly evaluated with Pareto analysis [45]. It follows that mathematical programming cannot provide diversified system operation. However, without the support provided by these techniques, it is necessary to develop algorithms that are capable of efficiently scheduling the assets. Furthermore, considering the actual cost of BESS, the profits from multiple applications are fundamental to generating a positive investment. Therefore, algorithms capable of effectively stacking the services are necessary to improve the economics of a storage system. Implementing those algorithms is not straightforward, since BESS are limited in power and energy. The desired output of the service stacking problem is a strategy for optimal capacity allocation during a given period, accounting for market prices and system dynamics. Three distinct types of service stacking are defined in the literature: sequential, parallel, and dynamic [46]. In parallel stacking, a constant allocation of storage capacity is given to the services whilst the sequential multi-use provides these services in turn. The dynamic multi-use aims to increase profit as it combines the advantages of the two predecessors. It follows that service stacking in sizing applications is a complex topic and aspects such as the type of stacking

5 of 24

or the services' optimum sizing represent important challenges for the stakeholder [47]. Furthermore, power plant configuration has a relevant role in the BESS-sizing procedure due to the different services that this technology can provide. For instance, a BESS coupled with an energy district could perform tasks such as load peak reductions and smooth power injections. Vice versa, in a stand-alone setup, the main activities are exclusively grid- and market-oriented. BESS-sizing approaches have been categorized into four main categories to deal with different configurations: microgrids, distributed renewable energy systems, standalone hybrid renewable energy systems, and renewable energy power plants [48]. Despite the wide interest in BESS-sizing in the literature, the proposed approaches typically focus on a single configuration. Conversely, industrial and utility-scale BESS stakeholders are interested in flexible tools that are able to evaluate heterogeneous configurations of power plants, storage systems, markets, and services.

Table 1 lists the studies that have addressed the stacking of services for a grid-scale BESS.

Reference	Services	Performance Assumptions	Configuration	Scope
[49]	Fast Frequency—Balancing market	Efficiency as a function of SOC and power	Stand-alone	Operation
[50]	Arbitrage—Frequency regulation	Constant efficiency	Stand-alone	Operation
[51]	Arbitrage—Frequency regulation	Constant efficiency	Stand-alone	Operation
[52]	Arbitrage—Frequency regulation-	Constant efficiency	Stand-alone	Sizing
[53]	Power shifting—aFRR	Constant efficiency	Wind farm coupling	Operation
[54]	Ancillary services market	Constant efficiency	Stand-alone	Operation
[55]	Arbitrage—distribution investment deferral—frequency regulation	Constant efficiency	Stand-alone	Operation
[34]	Arbitrage—Frequency regulation	Non-linear power losses depending on C-rate and SOC	Stand-alone	Operation
[56]	Frequency regulation—power shifting	Constant efficiency	Microgrid	Sizing
This work	Arbitrage—Frequency regulation—mFRR— capacity market	Efficiency as a function of SOC and power	Stand-alone and PV–wind coupled	Sizing

Table 1. List of studies that have inspected the stacking of services for a grid-scale BESS.

As stated in the Introduction, this work improves on the state-of-the-art by developing a high-fidelity model of BESS and novel stacking algorithms, considering both stand-alone and RES-coupled operations.

3. Methodology

This work proposes a novel flexible tool evaluating the size of a BESS and computing its cost-effectiveness while providing multiple services. The exploitation of the tool has been eased thanks to a graphic user interface (GUI) developed in *MATLABTM*. The structure allows for the easy evaluation of candidate groups of BESS and identifies the configuration that ensures the best economic return. The GUI compares different specific BESS sizes to accurately assess the trade-off between the size and the cost. The procedure is designed to be compatible with the generic EU market (e.g., it implements standard balancing products, and considers the day-ahead market according to the EU framework), meaning that it can be easily adapted to different requirements. However, the focus of the proposed algorithm in this paper is specifically related to the Italian market (e.g., the ancillary services market price scenarios and award rates, a well as the rules and prizes of the capacity remuneration mechanism, suit the Italian case).

This procedure may investigate both stand-alone and BESS hybrid-renewable energy power plants with wind and photovoltaic services connected to the national grid. Furthermore, the power managed at the point of delivery (POD) with the grid can be limited to a specified value to address study cases where there is an existing contractual connection power, and the curtailment of the overproduction is foreseen. The proxy model adopted in the procedure is a high-fidelity (cf. realistic) empirical battery model based on an experimental campaign at the Joint Research Centre (JRC) of Ispra (VA-Italy) on nickel-manganese cobalt (NMC) BESS [9]. The main features are the efficiency of the overall system, including transformer and power-conversion systems, and the auxiliary consumption. The nonlinear round-trip efficiency is expressed as a function of power and SOC through look-up tables. Instead, the auxiliaries' consumption relies on the power flown in the BESS and the ambient temperature. Although the model emulates the performances of NMC technology, its structures easily allow for the fitting of the data of different electrochemical technologies (e.g., lithium iron phosphate or even non-lithium-based batteries). Figure 1 depicts the block diagram of the high-fidelity empirical model adopted in this work using SIMULINKTM. Each block has a specific purpose that ensures the emulation of a large-scale BESS. The overall efficiency block takes the AC power, and the SOC as input and converts them into DC power using the previously mentioned look-up table. After this, the DC input is processed by the capability curve that limits the C-rate depending on the actual SOC. Lastly, the actual C-rate is used to update the SOC of the system. Inside the latter block, the energy content of the system is evaluated at each time sample.



Figure 1. Block diagram of the sizing tool developed in this work.

To ensure a realistic representation of BESS performances, an aging model from the literature was inserted into this framework. It is well-known that capacity degradation is affected by various chemical reactions, which can typically be classified into two major phenomena: calendar and cycle aging [57]. To account for these processes, two equations that describe the cycle and the calendar aging of the system have been inserted into the SOC update block to reduce the nominal capacity throughout the simulation. Both expressions were obtained from the literature. The cycle aging is related to the use of the BESS and its C-rate. The complete expression implemented in the model was obtained in [58].

Cycledegrad.[%] =
$$-[SOC(t) - SOC(t-1)] \times 3.57 \times 10^{-5} \times e^{0.465 \times C \cdot rate}$$
 (1)

Equation (1) describes the system degradation due to cycling by evaluating the depth of discharge between each time sample [SOC(t) - SOC(t - 1)] and the specific C-rate at which the BESS is cycled. The proposed tool investigates the BESS operation with a 15-min granularity, over a very long time window of up to several years; consequently, the equation

computes the percentage of capacity reduction generated in 15 min by a specific constant C-rate.

The second degradation mechanism has been modeled as a constant-capacity degradation in the square root of time function. This trend has been widely adopted to describe the chemical reaction inside the system that occurs due to calendar aging. The expression adopted in this work is based on [59], assuming an average SOC of 50% for the system.

Calendardegrad. =
$$-99.43 \times 10^3 \times e^{-\frac{42577}{RT}} \times \sqrt{t}$$
 (2)

Equation (2) reports the capacity calendar degradation depending on the age of the storage system, considering the gas constant R and the temperature T at which the system is kept by the auxiliaries. It has been assumed that the two effects superimpose to continuously update the SOH of the system. Therefore, at each time sample the BESS' nominal energy is reduced depending on the C-rate and the time that has elapsed.

The left side of Figure 1 reports the possible power plant setup evaluated by the tool. The inputs for the analysis are time series that can be ascribed into three different categories: energy, market, and ambient inputs. An additional block considering the optional presence of RES plants was also included to investigate hybrid configurations. Energy input returns the MW production of a given RES plant, if present. Market input contains fundamental information about the structure of each service being addressed. Lastly, temperature inputs are necessary for the auxiliary consumption of the BESS model. These parameters can be easily changed thanks to the GUI, which allows for the selection of the Excel file that is necessary for the analysis.

The evaluation of the nominal energy of the BESS depending on a given set of inputs is evaluated by the tool with iterations of different sizes through the definition of two distinct sets of parameters: nominal power and energy-to-power ratio (EPR). Moreover, if the storage system is coupled with RES, the tool can evaluate variable power plant sizes.

The cost-effectiveness of BESS relies on the profits that the asset can generate by participating in different services. Therefore, different algorithms were developed to emulate the participation of the BESS in various energy markets. Each algorithm elaborates the inputs and returns a power setpoint that simulates the dispatching of the BESS power flows or the possibility of charging using the assets in the energy district.

The algorithms proposed in this paper are analytical adaptive algorithms. Every day, the input data are fed to the algorithms that compute the operation of the BESS as a combination of charge and discharge signals. The simulation calculates the cash flow for twenty years for each size. The outcomes of each analysis are exploited to compute the performance indicators fundamentals to determine the optimal size of the system. The tool evaluates the internal rate of return (IRR) and the CAPEX that is covered to quantify the cost-effectiveness of each investment. The IRR was computed with the financial toolbox implemented in $MATLAB^{TM}$. Instead, the CAPEX covering 20 years was obtained following Equation (3), using the constant interest rate r.

$$CAPEX_{covered}[\%] = \frac{100}{CAPEX} \times \sum_{t=1}^{20} \frac{Revenues(t) - OPEX(t) - Marketpenalties}{(1+r)^{t}}$$
(3)

CAPEX and OPEX are properly addressed in the analysis. Equation (4) describes the capital cost as a product of the nominal energy and an exponential function of the duration. This expression was obtained through an interpolation of the values reported in [60].

$$CAPEX[\mathbf{\ell}] = E^{nom} \left(220 \times duration^{-0.9795} + 287.1 \right)$$
(4)

This condition is justified by the fact that equipment cost is strictly correlated with the power being handled and not only with the nominal energy. The hypothesis adopted for this work is that yearly OPEX corresponds to 2.5% of the CAPEX [61].

Lastly, due to the novelties of the application, a validation procedure based on previous works has not been added. Nevertheless, the high-fidelity empirical model represents an updated version of a previously developed structure. In [62], the adopted model has been compared with state-of-the-art models: the accuracy of performance representation increases when using a variable BESS efficiency and considering auxiliary system demand.

The algorithms implemented in the tools aim to emulate participation in different services. The next paragraphs briefly describe the logic adopted for their implementation. Four services typically present in EU markets have been modeled: energy arbitrage, capacity market participation, the provision of manual Frequency Restoration Reserve (mFRR), and the provision of fast frequency regulation within the Fast Reserve (FR) project. Each algorithm generates a signal that simulates the charge and discharge requirement for the BESS that participates in the specific service.

3.1. Energy Arbitrage

The term energy arbitrage refers to the possibility of storage systems exploiting dayahead market spread to generate profits. To achieve this control, the algorithm identifies the minimum and maximum prices of the market each day to set up the proper control logic. The algorithm was limited to a cycle per day to cope with the uncertainty of the prices that are always present in a multi-year analysis. To grant an economic profit, the procedure has a feedback control that checks whether the cycling cost of the system cancels the profits per cycle. Namely, this cost can be described as the total lifetime cost of the investment in electricity storage, divided by the cumulative delivered electricity. This parameter is customizable from the GUI and impacts the arbitrage provision. Furthermore, in case of BESS coupled with RES, if the RES power plant produces a power higher than the POD limit (i.e., the contractual connection power), the algorithm directs the exceeding generation toward the BESS if it is not fully charged. The latter charging process is considered free of charge since that energy will be curtailed if not stored. To properly address this aspect, an SOC control is implemented inside the algorithm to check the boundary condition of the storage system. Lastly, it has been hypothesized that the BESS acts as a price-taker in the day-ahead market. Algorithm 1 describes the structure of the energy arbitrage algorithm in detail.

3.2. Capacity Market

The electricity capacity market is coupled with the energy market. The scope of this service is to ensure power plant owners have sufficient capacity to meet the system demands and reliability. Participation in the market is defined through a tender procedure. The winners receive the capacity payment if they correctly provide power during the scarcity hours defined by the authority. In Italy, the assets that won the tender are entitled to the capacity payment if they participate in the day-ahead market in 1000 mandatory hours. The capacity payment differs from each power plant and is based on the probability of derating a given technology. Namely, the Italian authority associated a parameter with each type of power plant, related to their reliability in producing the nominal power [63]. This value reduces the capacity payment depending on the characteristics of each technology. For instance, BESS derating is shown in Table 2 and is proportional to the duration of the storage system, which means that a larger duration ensures better reliability in the provision of the capacity. Thus, the remunerated power of a BESS is its qualified power times (1—derating).

Algorithm 1: Energy Arbitrage

Input: P_i^{prod} , P^{nom} , E^{nom} , POD_{limit} , P_i^{DAM} , $SOC_0^{initial}$ Output: P_{cha}, P_{dis} CHARGE PHASE for I in 96 (quarter of hours in a day) do $if P_i^{prod} > POD_{limit}$ $i POD_{limit}$ $if P_i^{prod} - POD_{limit} < P^{nom}$ $P_i^{cha} = P_i^{prod} - POD_{limit}$ Else $P_i^{cha} = P_{nom}$ $E^{cha} = E^{cha} + \frac{P_i^{cha}}{4}$ If $sum(E_i^{cha}) < E^{nom}$ $j = find\left(min\left(P_i^{MGP}\right)\right)$ for i in 96 (quarter of hours in a day) do if i == j $if P_i^{prod} < P^{nom}$ $P_i^{cha} = P_i^{prod}$ Else $P_i^{cha} = P^{nom}$ $E^{cha} = E^{cha} + \frac{P_i^{cha}}{4}$ **DISCHARGE PHASE** $E^{dis} = E^{cha}$ $j = find \left(max \left(P_i^{MGP}\right)\right)$ for i in 96 (quarter of hours in a day) do while $E^{dis} > 0$ $\mathbf{if} \ i == j$ if $P_i^{nom} + P_i^{prod} > POD_{limit}$ $P_{i}^{dis} = POD_{limit} - P_{i}^{prod}$ Else $P_i^{dis} = P^{nom}$ $E^{dis} = E^{dis} - \frac{P_i^{dis}}{4}$ SOC CONTROL for i in 96 (quarter of hours in a day) do $SOC_i = SOC_{i-1} - \frac{P_i^{dis}}{4} + \frac{P_i^{cha}}{4}$ If $SOC_i > 100$ $\begin{array}{l} SOC_i = 100 \\ SOC_i = 100 \\ P_i^{cha} = \frac{100 - SOC_{i-1}}{4^* E^{nom}} \\ \text{Else if } SOC_i < 0 \end{array}$ $\begin{array}{l} \text{SOC}_i = 0 \\ P_i^{\text{cha}} = \frac{\text{SOC}_{i-1}}{4^* \text{E}^{\text{nom}}} \end{array}$ PROFITS PER CYCLE CONTROL for i in 96 (quarter of hours in a day) do If $P_i^{prod} > POD^{limit}$ $P^{chargefree} = min \Big(P^{nom}, P_i^{prod} - POD^{limit} \Big)$ $P^{chargepaid} = P_i^{charge} - P^{chargefree}$ $profit(i) = \left(0.9 \times P_i^{discharge} - P^{chargepaid}\right) \times \frac{DAM_i}{4}$ If sum $\left(\text{profit} - \text{Profits}^{\text{Cycle}} / E^{\text{nom}} \right) \le 0$ $\begin{array}{l} P^{discharge}_{1:96}=0\\ P^{charge}_{1:96}=0 \end{array}$

Duration [h]	Derating [%]		
1	76		
2	66		
4	33		
6	19		
8	10		

Table 2. Storage derating depending on the duration of the capacity payment in the capacity market defined by the Italian authority.

Furthermore, the non-programmable RES, due to their reliance on natural phenomena, face difficulties in providing a constant power setpoint. Hereby, the Italian authority has defined a derating factor for the capacity payment that these power plants can receive. Table 3 details the derating factor for the capacity payment of wind and PV power plants.

Table 3. Power plant derating factor defined by the Italian authority for the capacity payment.

Technology	Derating [%]	
PV	84	
Wind	88	

In this work, when the capacity algorithm is enabled, it is assumed that the hybrid power plant has won the tender for, and participates in, the capacity market. The capacity payment can be extended to account for a hybrid system with multiple energy sources following Equation (5), where the overall remunerated capacity *CP* is equal to the summation of the product between the peak power $P_{powerplant}^{peak}$ and the derating $Derating_{powerplant}$ of each technology multiplied by the capacity payment Capacity^{MW}_{payment} specified by the user. Algorithm 2 describes the logic used to emulate the capacity market inside the model.

$$CP[\mathbf{\epsilon}] = \sum_{powerplant=1}^{N} P_{powerplant}^{peak} \times \left(1 - Derating_{powerplant}\right) \times Capacity_{payment}^{MW}$$
(5)

Algorithm 2: Capacity Market Algorithm

The derating parameters adopted in this work are those proposed by the Italian authority and reported in Tables 2 and 3. Nevertheless, the tool allows for those values to be adapted to a different regulatory scenario.

Lastly, the inadequacy of an asset to properly participate in the capacity market is computed ex-post. This calculation is mandatory to verify if the system correctly provides the requested capacity and is entitled to the capacity payment. The criteria established by the Italian authority foresee a neglection of the payment when the power plant does not provide more than 80% of the capacity for at least three months.

3.3. Manual Frequency Restoration Reserve (mFRR)

mFRR is the manual activation of frequency reserve that has the purpose of restoring the power balance of the electric power system. In Italy, this service is traded in the ancillary services market with a pay-as-bid approach. In [64], it has been pointed out that distrusted energy sources such as hybrid RES plants have a very low share of acceptance in the market. Despite the small liquidity of the ancillary services market, the participation of the asset in these services could be profitable. Indeed, as an inherent structure of the process, the mFRR prices tend to be less competitive than those in the day-ahead market and consequently generate more profit. To properly model the opportunity generated by participation in the mFRR, an acceptance criterion has been developed in [65]. The procedure, based on historical results on mFRR, develops around two important pieces of information. Firstly, among all the bids that were submitted, only 13% were accepted. Secondly, a correlation between bid acceptance and the submission price is present. In light of these statements, the tool was programmed to create a binary input vector that expresses the acceptance of the bids in the mFRR. The acceptance array changes depending on the average price that the user foresees being submitted in the market. Lastly, the accepted bids are remunerated at a value that is equal to the constant price times a gain that is proportional to the hour at which the bid is accepted. This solution was implemented to grasp the hourly price fluctuations in the bids in the ancillary services market.

Algorithm 3 describes the logic behind the control of the system participating in the mFRR. In particular, the approach evaluates whether the mFRR bids are more profitable than those in the day-ahead market to ensure higher revenues.

Algorithm 3: mFRR Algorithm

```
\begin{split} \textbf{Input:} \ P_i^{prod}, P^{nom}, E^{nom}, POD_{limit}, P_i^{MGP}, P_i^{mFRR} \\ \textbf{Output:} \ P_{cha}, P_{dis} \\ \underline{\textbf{DISCHARGE PHASE}} \\ \textbf{for I in 96 (quarter of hours in a day) do} \\ \textbf{if sum}(P_i^{mFRR}) > 0 \\ \textbf{if } j = find\left(max\left(P_i^{mFRR}\right)\right) \\ \textbf{if } p_j^{mSD} > P_j^{MGP} \\ \textbf{if } P_j^{prod} > POD_{limit} \\ P_j^{dis} = 0 \\ \textbf{Else if } POD_{limit} - P_j^{prod} > P^{nom} \\ P_j^{dis} = P^{nom} \\ \textbf{Else} \\ P_j^{dis} = POD_{limit} - P_j^{prod} \\ E^{dis} = E^{dis} - \frac{P_j^{dis}}{4} \end{split}
```

3.4. Fast Frequency Reserve

Fast frequency reserve is a service enabled by the Italian authority with the resolution 200/2020/R/eel [66]. It consists of the rapid provision of power from BESS to counteract the more usual frequency swings in the electric power system. The service is exclusive to

BESS selected through a tender procedure. In detail, the pilot project launched in 2021 in Italy mandated the provision of a fast frequency reserve for 1000 specified hours for the storage systems entitled to participate in this market. In those periods, the assets must ensure the provision of the power submitted during the tender to the service and grant it in all the hours specified by the authority. The correct interaction of the BESS with the grid is remunerated with a capacity payment from the TSO that is proportional to the power made available to the service. Fast frequency reserve requirements limit both the power and energy of the storage system. Algorithm 4 describes the fast frequency reserve algorithm as a limitation of the available power and energy during the hour mandated by the authority for the service.

Algorithm 4: Fast Frequency Reserve

Input: P^{prod}, P^{nom}, E^{nom}, POD_{limit}, P^{MGP}, P^{mFRR}, P^{qualified}, P^{FRU}, FRU^{signal} **Output:** P_{cha}, P_{dis} SOC Control for i in 96 (quarter of hours in a day) do
$$\begin{split} \text{SOC}_{i}^{\text{max}} &= 100 - \left(\frac{\text{Pqualified}}{4 \times \text{E}^{\text{nom}}} \times 100\right) \times \text{FRU}^{\text{signal}} \\ \text{SOC}_{i}^{\text{min}} &= \left(\frac{\text{Pqualified}}{4 \times \text{E}^{\text{nom}}} \times 100\right) \times \text{FRU}^{\text{signal}} \end{split}$$
If $P_{i}^{cha} > P^{nom} - P^{FRU}$ and $FRU^{signal} = 1$ $P_{i}^{cha} = P^{nom} - P^{FRU}$ **Else if** $P_i^{dis} > P^{nom} - P^{FRU}$ $P_{i}^{dis} = P^{nom} - P^{FRU}$ $If \ SOC_i > SOC_i^{max}$ $SOC_i = SOC_i^{max}$ $P_{i}^{dis} = \frac{SOC_{i}^{max} - SOC_{i-1}}{4 \times E^{nom}}$ Else if $SOC_i < SOC_i^{min}$ $SOC_i = SOC_i^{min}$ $P_i^{cha} = \frac{SOC_{i-1} - SOC_i^{min}}{4 \times E^{nom}}$ $SOC_i = SOC_{i-1} - \frac{P_i^{dis}}{4} + \frac{P_i^{cha}}{4}$

4. Study Cases

The sizing procedure was divided into two different study cases, labeled as standalone and RES-coupled. The purpose of this classification is to highlight the different benefits and shortcomings of the two configurations and the tool's ability to operate in different configurations. The stand-alone case analyzes the performance of the sole BESS interacting with the grid. The second one foresees a hybrid RES power plant coupled with a BESS that is capable of exchanging energy with the grid and the RES. In each study case, four different sets of simulations were carried out with different combinations of the algorithms, as described in Section 3. Each set of simulations aims to properly allocate different combinations of services to investigate the impact that a multi-use BESS has on its economics. The first simulation exclusively addresses the provision of energy arbitrage by the storage system. The second set couples the first service that was investigated with the capacity market through sequential stacking. In detail, the capacity algorithm has the dispatch priority due to the necessity of satisfying the energy provision mandated by the service regulations. The third simulation package couples the mFRR with the arbitrage and capacity algorithms. The ancillary service structure interacts with the arbitrage to identify the most profitable trading strategy for each day. It follows that the most remunerative dispatch is selected depending on the day-ahead market (DAM) prices and the acceptance in the mFRR. Lastly, the fast frequency reserve is evaluated in the fourth set of simulations stacked with the other services. The provision of this application is superimposed with the previous markets that were modeled. As a consequence, the fast reserve and a further service can be provided in parallel, and the latter is derated by the power and capacity required for the frequency regulation. The list of simulations described has allowed for

an investigation of the impact that service stacking has on the revenues of the BESS. The tool iterates the power and EPR of the system and identifies the optimal size of different configurations using the CAPEX being covered and IRR. Figure 2 summarizes the sets selected for each study case. Furthermore, the figure specifies the stacking classification of the investigated service. The number of services included in the stacking configuration is incremental. The first one investigates sole arbitrage. Configuration 2 adds the capacity market to the arbitrage. The third one involves the mFRR in the previous services. Finally, configuration 4 investigates the stacking of all the services.



Figure 2. Summary of the study case and the service stacking proposed for this study.

It is worth stating that the parameters reported in Figure 2 are completely customizable through the GUI to cover different setups. The tool evaluates the economic performance of the BESS, allowing for the charge from the grid or the power plants.

Moreover, a POD limit was established to limit the power injection into the grid. Table 4 details the parameters adopted for the study cases. The capacity payment duration complies with the current Italian framework [67]. Furthermore, the fast reserve being implemented reflects the actual one in place in Italy [68]. Both capacity payments refer to the outcomes of the auction for the Sardinia market zone. To correctly compare the latter service with BESS of various sizes, the power enslaved to the fast reserve is defined as a percentage of the nominal power of the system.

Table 4. Parameters adopted in the study case presented in this work.

Parameter	Stand-Alone	RES-Coupled
Power [MW]	[10, 20, 30, 40, 50]	[10, 20, 30, 40, 50]
Duration [h]	[3, 4, 5, 6]	[3, 4, 5, 6]
Wind peak power [MWp]	none	[20, 30, 40]
PV peak power [MWp]	none	[20, 30, 40]
POD limit [MW]	none	[40]
Temperature	Sardinia Italy, 2021	Sardinia Italy, 2021
Prices [years]	2019-2022	2019-2022
Capacity market duration [years]	15	15
Fast reserve duration [years]	3	3
Capacity market payment [$\varepsilon \times$ year/MW]	51,012	51,012
Fast Reserve payment [€ × year/MW]	64,890	64,890
Fast Reserve Power [%]	20% P _{nom}	20% P _{nom}
Interest rate [%]	6	6
Profits per cycle [€/MWh]	25	25

The input vectors describe a twenty-year time series with a time sample of 15 min. The main input vectors are reported in Figure 3. The wind and PV profiles were generated by the open-access software AtlanteEolico-RSE [69] and PVGIS, respectively, for Southern Italy. The lack of seasonality of the wind was modeled by randomly sampling the weeks of the three years wind profile used as input. The PV power profile was reduced every year by a degradation equal to 1%/year. The historical temperatures were downloaded from the ARPA website for Southern Italy and linearly increased up to 2 °C to account for global warming [70]. Lastly, the day-ahead market prices of the year 2019 were exploited for the study case [71]. The profile was chosen because it represents the last year of business as usual before the pandemic and the gas shortages. Furthermore, due to the high prices recorded in 2021 and 2022, the 2019 price profile can be adopted as a conservative solution for the investment cash flow. A one-year simulation exclusively involving the arbitrage with the 2019-2020-2021 prices profile was developed, resulting in different realized revenues, as depicted in Table 5.







Table 5. Revenues for one year of arbitrage made by a 20 MW/60 MWh BESS with three different price scenarios.

Price Scenario	Revenues [k€]	CAPEX Covered in the First Year [%]
DAM 2019	388.77	1.82
DAM 2021	762.22	3.58
DAM 2022	2125.54	9.98

The scope of this brief estimation has been the identification of the price profile that grants a lower economic return for the investment and can be used as input for the main analysis. The considered trend was extended to the period under evaluation and coupled with two different yearly gains. The scope of the gains is to create two distinct price scenarios to cope with the possible evolution of the prices in the future and provide a wider economic analysis of the investment. The yearly gains, defined as low- and high-price scenarios, were extrapolated from the results reported in [72]. The profits per cycle were kept low to understand the main differences that arise in the BESS cycling due to this parameter.

5. Results

This section presents the results obtained from the different algorithms implemented in the sizing tool. The results focus strictly on the economic indexes that are fundamental for evaluating the profitability of an investment. For the sake of completeness, both the covered CAPEX and IRR are reported. Furthermore, technical pieces of information are provided through the full equivalent cycle (FEC) and the capacity fades of the storage system under analysis.

Firstly, it is worth stating that the tool takes from 4.3 to 12.1 min (Intel i7 1260 CPU— 16 GB) to simulate 20 years of service of a sole BESS with the lightest and heaviest algorithm; such a limited computational effort validates the approach's viability in tecno-economical studies. In the specified period, the procedure evaluates the power flow, degradation, and revenues of the BESS to provide valuable information on the investment.

The first reported outcomes concern the identification of a price profile that can act as a conservative solution. Table 5 details the revenues generated in one year by a 20 MW/60 MWh BESS, which performs energy arbitrage with three different price scenarios: the day-ahead market prices of 2019, 2021, and 2022.

It is possible to note that the larger volatility of the prices during the gas shortages returns better economics for the storage system. In particular, almost 10% of the 21.3 M€ of BESS CAPEX is covered in a single year in the most profitable price scenario. Nevertheless, these profits were generated by the abnormal market conditions of 2022 that, thanks to the more widespread adoption of RES, will not recur in the coming years. It follows that DAM prices for 2019 were exploited as a base to develop high and low price scenarios for the next 20 years.

The algorithms have different impacts on the economics of the system. The aim of the following paragraphs is to detail the benefits of each approach and their limitations in both stand-alone and RES coupled configurations to understand the outcomes of the tool in more depth.

5.1. Study Case 1: Stand-Alone BESS

5.1.1. Stacking Configuration 1: Energy Arbitrage

The arbitrage algorithm investigated in stacking configuration 1 is by far the less remunerative scheme for a stand-alone system. Firstly, the profit is strictly related to the spot market outcomes and, generally, a certain degree of uncertainty characterizes these variables. Furthermore, the operation of the BESS is connected to the market price spread. The high-price scenario has a larger spread than the low-price profile; consequently, it represents the most remunerative profile. The IRR difference between the same BESS with the two price trends corresponds to 4.5% on average. This value corresponds to a difference in revenues between the two scenario prices equal to 12.0% of the CAPEX for the whole investment lifespan. Furthermore, it is important to correctly estimate the profits per cycle to seize the correct trade-off between the operation of the system and the covered CAPEX. A reduction in the capital cost will directly impact the cycling cost of the system and increase the possibility of profits for arbitrage. The degradation is strictly related to these parameters, since a larger cycling will correspond to a higher cycle aging. The final capacity after 20 years for the arbitrage service is, on average, equal to 81.3%, considering a periodic capacity augmentation. Revenues are linearly impacted by the degradation. The phenomenon leads to a reduction in the profit at the end of the investment equal to 12.2%. A possible solution to increase the revenues from the arbitrage algorithm consists of increasing the cycling limits of the BESS set to one cycle per day. This condition may seem quite conservative for a system with a small EPR. Indeed, it is possible to increase profits if the market price spread allows for multiple cycles. However, profits larger than the profits per cycle must be always granted to perform the arbitrage. This condition is difficult to ensure with a higher number of cycles because the spread between the charge and discharge must be higher than the cycling cost. Furthermore, for a large EPR (>5 h), the benefits of multiple cycles are null due to the inherent structure of the market outcomes, which, in

general, has two peaks lasting 2–3 h each [73]. It follows that BESS with durations over 5 h are less-suited to the daily energy arbitrage. As a consequence, the best configuration for the sole arbitrage is a 3 h BESS that correctly grasps the variation in the day-ahead market. Lastly, the revenues for BESS with the same EPR are linear with power because the efficiency of the service is exclusively related to the duration of the storage system. Therefore, BESS with the same duration, which mostly provide the arbitrage at the nominal power, have a similar IRR.

5.1.2. Stacking Configuration 2: Energy Arbitrage + Capacity Market

The capacity market represents an important source of income for the storage system as evaluated in configuration 2. The fixed revenues ensured by the provision of the services increase the economic return of the asset. However, in the best scenario, the profits cover only 55.8% of the CAPEX. Nevertheless, a stand-alone system must participate in this service to cover the missing money due to the excessive capital cost. Moreover, the service does not increase the degradation compared to the sole arbitrage strategy: indeed, in Italy, capacity remuneration is due if the obligation of bidding the remunerated capacity on the market is respected, as BESS would do in the case of energy arbitrage. However, capacity fade is slightly lower since the algorithms limit the BESS cycling to ensure the provision of the energy required by the regulation. Furthermore, due to the structure of the Italian capacity market, a longer duration of BESS benefits more from the capacity market. Indeed, as reported in Table 2 a lower derating factor is mandated by the authority of storage systems with a longer duration. This condition coupled with the economy of scale granted by a larger system EPR, which identifies BESS with a 5 h duration as the best candidate for the provision of arbitrage and capacity services.

5.1.3. Stacking Configuration 3: Energy Arbitrage + Capacity Market + mFRR

The addition of the mFRR to the services provided in stacking configuration 3 grants slightly better revenues thanks to the higher profits generated by the bids in the ancillary services market. On average, the increase in profits attests to around 1–2% of the CAPEX for the whole investment lifespan. In this case, the limits are mainly driven by the low award rate of assets on the Italian ancillary services market. However, with the increased penetration of RES, the liquidity of these assets inside the ancillary services market could increase. Therefore, future updates based on new market analysis may lead to an important increase in the cash flows. In the upper part of Figure 4, it is possible to observe how the BESS is dispatched by Algorithm 3, which selects the most profitable solution between mFRR and DAM. Furthermore, an SOC control, as depicted in the bottom part of the image, ensures that the BESS can efficiently participate in multiple services.



Algorithm 3 with a stand-alone 20 MW/60 MWh BESS

Figure 4. Algorithm 3 output for a stand-alone 20 MW/60 MWh BESS.

5.1.4. Stacking Configuration 4: Energy Arbitrage + Capacity Market + mFRR + FR

Lastly, the fast frequency reserve algorithm has been coupled with the other three in the stacking configuration 4. It has been noted that the impact of these initiatives on the economics of the system is inversely proportional to the nominal energy of the asset. Indeed, the regulation foresees a capacity payment based on the percentage of the nominal power enslaved to the service. However, the CAPEX of the asset increases with the duration of the BESS. Therefore, the service leads to a higher improvement in the economics of systems with a low duration. The covered CAPEX, thanks to three years of fast reserve service, goes from 2.0% of the 6 h BESS to 3.5% of the 3 h system. It is worth stating that the revenues generated by the capacity market maintain the 5 h duration system as the best investment for the stand-alone configuration. Nevertheless, the hypothesis adopted for the fast reserve is conservative. Indeed, after the three years mandated by the authority, no further service is foreseen for fast reserve provision inside the tool. However, the fast reserve provided by BESS may be standardized in the future and remunerated with regulations that are not yet specified. Table 6 lists the best results for the stand-alone study case for each stacking configuration. It is possible to observe that, despite being negative in all stacking configurations under our assumptions, the best IRR is granted by configuration 4, which is the solution that couples the higher number of services. Nevertheless, it is worth noting that the adopted price scenarios are conservative and return the minimum profit that BESS can generate. Furthermore, better economics are not strictly related to the asset cycling if capacity payments are granted to the storage system. Indeed, the same number of cycles allows for the revenues of the storage system to be tripled thanks to its participation in the capacity market and fast reserve provision. The general outcome of the stand-alone configuration is that, for a one-cycle-per-day system, the 20-year investment is not remunerative, no matter the adopted revenue stacking strategy: this unprofitable result is mainly due to the conservative assumption of using the 2019 DAM price scenario. However, the coupling of various services boosts the CAPEX covered by the profits from 49% to 60%. Therefore, the asset operator must carefully evaluate which services the system can provide to increase profit and ensure capacity payment. This is fundamental, especially for those services where participation is granted through an auction mechanism. Indeed, thanks to the capacity payment, the best investment is represented by 10 MW/50 MWhBESS. In general, systems with a duration of 5 h have better economics because they represent the optimal trade-off between capacity payment and CAPEX.

Stacking Configuration	Best BESS Sizing	IRR [%]	CAPEX Covered [%]	FEC	Energy Fade [%]
1	10 MW/30 MWh	-1.72	49	3183.3	18.68
2	10 MW/50 MWh	-1.03	55	3104.7	18.05
3	10 MW/50 MWh	-1.00	57	3141.1	18.40
4	10 MW/50 MWh	-0.95	60	3141.1	18.51

Table 6. Main results for the stand-alone study case (study case 1).

5.2. Study Case 2: RES-Coupled BESS

5.2.1. Stacking Configuration 1: Energy Arbitrage

The second study case exploits the presence of RES to increase profits. Indeed, the energy curtailment mandated by the POD limit is equivalent to free energy that will be lost without the storage system. This parameter, together with the PV and wind size, show a large sensitivity regarding the free charging energy for the BESS and the energy lost by the power plant. Although the overproduction of RES provides advantages to the BESS, it is fundamental to highlight that the configurations with the lowest peak power production barely improve the economics of the BESS. In the worst case, namely, a covered CAPEX of 50 MW/300 MWh BESS this increases by only 3% following the addition of 20 MWp of both PV and wind compared to the stand-alone case. A further observation related to the

coupling of the asset to the RES is the higher number of cycles. The availability of cheap energy for charging the system due to the presence of RES increases the use of the BESS and its degradation, which reaches over 30% for the system with a smaller duration. However, increased use of the storage system corresponds to better economics; as augmentation of the asset is usually included in the OPEX, it is always better to cycle the system and boost the profits.

The presence of a power plant allows for the BESS to be charged for free and save energy that, without the storage system, will be curtailed. This condition leads to benefits proportional to the energy being stored. In the best scenarios, namely high prices and high-power plant peak power, the investment of a BESS generates value with the sole arbitrage. The revenues from the sole arbitrage are almost doubled in the best-case scenario. The most cost-effective BESS size with the provision of the sole arbitrage is 10 MW/30 MWh. This system represents an optimal trade-off between the CAPEX and the ability to collect the byproducts caused by RES plant overproduction. Higher powers reduce the curtailment but not linearly. Therefore, the expenditure increase is higher than the revenue generated for the increment in energy savings. Instead, the EPR increases only the expenditures without further energy savings, leading to a higher cost and similar profit.

5.2.2. Stacking Configuration 2: Energy Arbitrage + Capacity Market

Stacking configuration 2, which couples the capacity market with the arbitrage, improves the economic results. Following Equation (5), coupling with RES led to a highercapacity payment compared to the stand-alone case. This condition drastically improves the economics of the system. However, this increase in capacity may lead to a mandatory power provision value that is larger than the nominal power of the BESS. Therefore, if the unpredictable production of RES goes to zero, in some cases, the BESS does not have the necessary characteristics to fulfill the constraints of the capacity market. It follows that, for a small BESS and large RES-size capacity, payment is not received due to the power limits of the storage system. However, larger BESS benefit from this condition, with an improvement of up to 30% in the CAPEX covered compared to the sole arbitrage provision. This aspect is fundamental to justify the presence of a storage system in a hybrid energy district. Indeed, for small-scale BESS, the capacity payment is not granted and the same will occur if no BESS is present in the energy district. Therefore, for an RES owner that is willing to participate in the capacity market, it is mandatory to purchase a BESS to fulfill the requirements of the Italian authorities. Lastly, as in the previous study, participation in this new service has a very low impact on cycling and degradation. Figure 5 depicts the covered CAPEX for the RES-coupled study case with Algorithm 2. The heatmap produced by the GUI can identify the most cost-effective BESS for the scenario under analysis, namely the 20 MW/60 MWh system.

5.2.3. Stacking Configuration 3 and 4: Energy Arbitrage + Capacity Market + mFRR + FR

Lastly, mFRR and fast reserve provision has a similar influence on the RES-coupled study case to the stand-alone. Indeed, the two services are strictly related to the operation of the BESS rather than the presence of RES power plants.

The best results for the RES-coupled study case are listed in Table 7. The outcomes of the second study case are more impacted by service stacking than the first analysis. Indeed, the capacity prize is larger due to the necessity of coupling RES with a BESS to provide a constant power setpoint. The service stacking improves the covered CAPEX by up to 30% compared to the sole arbitrage provision. Furthermore, as stated in [74], BESS have an important economic advantage in supporting RES systems compared to the stand-alone case. The coupling with power sources reduces the expenditures on the energy sale on the day-ahead market and stabilizes the unpredictability of RES granting the capacity payment.



20 MW/60 MWh

CAPEX covered with Algorithm 2 in the RES-coupled study case (40 MWP wind ; 40 MWp PV)

Figure 5. CAPEX covering 20 years with different sizes of BESS for the RES-coupled study case with 40 MWp wind and PV.

 Stacking Configuration	Best Configuration	IRR [%]	CAPEX Covered [%]	FEC	Capacity Fade [%]
1	10 MW/30 MWh	5.97	99	6410.3	32.45
2	20 MW/60 MWh	10.46	138	6233.3	30.23
3	20 MW/60 MWh	10.55	139	6180.5	30.17

10.84

144

6124.2

29.87

Table 7. Main results for RES-coupled study case 2 with 40 MWp for both wind and PV.

6. Conclusions

4

This article proposes a model-aware analysis to resolve the BESS sizing issue considering different applications that implement service stacking. The economic analysis evaluated the investment throughout a 20-year scenario, exploiting a high-fidelity empirical BESS model equipped with equations that are able to describe both (i) the performance of BESS in different operating and ambient conditions and (ii) the capacity degradation of the asset. Novel algorithms were developed to properly emulate the BESS control strategy participating in the capacity and energy markets. Flexibility and reduced computational effort make the tool valid for the rapid assessment of different system configurations (e.g., stand-alone or RES-integrated), the sizing (e.g., the varying power and duration of the BESS), and the services (i.e., a control strategy block can implement complex algorithms to emulate service provision under different regulatory frameworks).

In this work, two systems were simulated, featuring a stand-alone and RES-coupled BESS, which served as case studies. The provided services are implemented to fit Italian regulation and market conditions. The outcomes highlighted the wide difference between the stand-alone and RES-coupled BESS' return of the investment. A stand-alone system that is cycled once per day could not cover more than 60% of the CAPEX in 20 years. Although the inclusion of capacity payments led to an improvement in terms of economics, the current Italian regulations do not provide sufficient revenue to return on the investment, as per the assumptions made in the study. However, in the future, reductions in the system

inertia will require more BESS use for frequency regulation. As a consequence, a new market or an extension of the existing mechanisms (e.g., fast reserve, limited to 1000 h per year as of 2023) could enter the pitch and provide new remuneration schemes for BESS. Conversely, the configuration presenting a BESS coupled with RES shows a positive IRR thanks to the optimal synergy between the storage and the intermittent power production. A comparison between the two study cases highlighted the advantage that BESS has in supporting the RES system.

This can be used to obtain the best result from the energy sale on the day-ahead market and allows for a better integration of the RES + BESS system in the grid. This is testified by the decreased connection power required for the hybrid RES + BESS plant and the exploitation of a larger capacity payment.

Optimal sizing is fundamental to obtaining a positive economic outcome, as highlighted by the results. Moreover, the system configuration play a key role in the economics of the asset. Indeed, the same services and the same power rating with RES can provide a 2.3 times higher return on the investment compared to the stand-alone case.

Additionally, the outcomes highlighted the necessity of service stacking to fully cover the investment costs. In particular, it has been pointed out that the sole energy service (i.e., arbitrage) cannot cover the capital costs of the assets, increasing the interest for BESS participation in capacity remuneration mechanisms or long-term contracts to ensure flexibility. The most efficient service stacking led to an increase in the covered CAPEX that was equal to 11% in the stand-alone study case and an astonishing 45% in the hybrid configuration. This improvement was achieved without increasing the number of cycles performed by the BESS.

The limitations of the study include the fact that the iterative procedure is not an optimization; thus, the optimal solution can be hidden between two tested configurations. Given the possibility of increasing the granularity of the simulations thanks to the low computational effort required, this is not considered a major issue. Additionally, the study was performed using a BESS model from the literature, featuring a Li-NMC battery. It is well-known that, as of 2023, LFP chemistry has overcome Li-NMC, especially in utility-scale applications. The generalization of the procedure to include an LFP battery model is of great interest.

Future works could focus on improving the algorithms proposed in this article. These include the possibility of cycling the BESS more than once per day (to better test the trade-off between BESS life and NPV), the inclusion of an aging-aware algorithm to identify an efficient trade-off between profit and degradation, and the testing of innovative ancillary services that can fulfil the new needs of power systems. Furthermore, the possibility of easily evaluating different BESS technologies will drive the comparison and identification of efficient services for a specific cell chemistry.

Author Contributions: Conceptualization, G.G., M.M., S.M. and G.R.; methodology, F.B. (Filippo Bovera),G.R., M.M. and M.S.; software M.S.; Validation F.B. (Federico Bresciani) and M.S., writing—original draft, M.S.; writing—review and editing G.R., F.B. (Federico Bresciani) and M.M.; visualization M.S.; supervision, M.M., G.G. and S.M.; project administration M.M. and G.G. All authors have read and agreed to the published version of the manuscript.

Funding: This work has been funded by Eni in the context of the Joint Research Center between Eni and Politecnico di Milano (Prot. Nr. 4310481954—Project B3M).

Data Availability Statement: Data will be made available on request.

Conflicts of Interest: The authors declare no conflict of interest.

Nomenclature

P ^{qualified} : Power qualified to provide FRU
P ^{FRU} : Power enslaved to FRU provision
FRU ^{signal} : FRU participation signal
Decision Variables
P _{cha} : BESS charging power
P _{dis} : BESS discharging power
E ^{cha} : Energy associated with the daily charge
E ^{dis} : Energy associated with the daily discharge
P ^{chargefree} : Power without charge from the hybrid energy district
P ^{chargepaid} : Power bought from the DAM
profit: Profits daily generated by the BESS

References

- 1. Horowitz, C.A. Paris Agreement. *Materials* **2016**, *55*, 740–755. [CrossRef]
- 2. Robert, C. Pietzcker, Sebastian Osorio, Renato Rodrigues, Tightening EU ETS targets in line with the European Green Deal: Impacts on the decarbonization of the EU power sector. *Appl. Energy* **2021**, *293*, 116914. [CrossRef]
- 3. European Commission. The European Green Deal. Bruxelles. June 2021. Available online: https://sustainabledevelopment.un. org/post2015/transformingourworld (accessed on 18 April 2023).
- 4. Antonelli, M.; Desideri, U.; Franco, A. Effects of large scale penetration of renewables: The Italian case in the years 2008–2015. In *Renewable and Sustainable Energy Reviews*; Elsevier Ltd.: Amsterdam, The Netherlands, 2018; Volume 81, pp. 3090–3100. [CrossRef]
- Bloomberg; EATON; Statkraft. Flexibility Solutions for High-Renewable Energy Systems. 2018. Available online: https://data.bloomberglp.com/professional/sites/24/2018/11/UK-Flexibility-Solutions-for-High-Renewable-Energy-Systems-2018-BNEF-Eaton-Statkraft.pdf (accessed on 5 June 2023).
- 6. International Energy Agency. World Energy Outlook 2022. 2022. Available online: www.iea.org/t&c/ (accessed on 20 May 2023).
- 7. Dunn, B.; Kamath, H.; Tarascon, J.-M. Electrical Energy Storage for the Grid: A Battery of Choices System Power Ratings, Module Size. Available online: www.sciencemag.org (accessed on 20 March 2023).
- 8. Englberger, S.; Jossen, A.; Hesse, H. Unlocking the Potential of Battery Storage with the Dynamic Stacking of Multiple Applications. *Cell Rep. Phys. Sci.* 2020, 1, 100238. [CrossRef]
- 9. Rancilio, G.; Lucas, A.; Kotsakis, E.; Fulli, G.; Merlo, M.; Delfanti, M.; Masera, M. Modeling a large-scale battery energy storage system for power grid application analysis. *Energies* **2019**, *12*, 3312. [CrossRef]
- Vykhodtsev, A.V.; Jang, D.; Wang, Q.; Rosehart, W.; Zareipour, H. A review of modelling approaches to characterize lithium-ion battery energy storage systems in techno-economic analyses of power systems. *Renew. Sustain. Energy Rev.* 2022, 166, 112584. [CrossRef]
- 11. Tamilselvi, S.; Gunasundari, S.; Karuppiah, N.; Razak RK, A.; Madhusudan, S.; Nagarajan, V.M.; Sathish, T.; Shamim, M.Z.M.; Saleel, C.A.; Afzal, A. A review on battery modelling techniques. *Sustainability* **2021**, *13*, 42. [CrossRef]
- 12. Moura, S.J.; Chaturvedi, N.A. Adaptive PDE Observer for Battery SOC/SOH Estimation via an Electrochemical Model. *Am. Soc. Mech. Eng.* **2012**, *45295*, 101–110.
- 13. Doyle, M.; Newman, J. The use of mathematical modeling in the design of lithium polymer battery system. *Electrochim. Acta* **1995**, 40, 2191–2196. [CrossRef]
- 14. Xu, L.; Lin, X.; Xie, Y.; Hu, X. Enabling high-fidelity electrochemical P2D modeling of lithium-ion batteries via fast and nondestructive parameter identification. *Energy Storage Mater.* **2022**, *45*, 952–968. [CrossRef]
- 15. Allam, A.; Onori, S. Online Capacity Estimation for Lithium-Ion Battery Cells via an Electrochemical Model-Based Adaptive Interconnected Observer. *IEEE Trans. Control Syst. Technol.* **2020**, *29*, 1636–1651. [CrossRef]
- 16. He, W.; Pecht, M.; Flynn, D.; Dinmohammadi, F. A physics-based electrochemical model for lithium-ion battery state-of-charge estimation solved by an optimised projection-based method and moving-window filtering. *Energies* **2018**, *11*, 2120. [CrossRef]
- 17. Barzacchi, L.; Lagnoni, M.; Di Rienzo, R.; Bertei, A.; Baronti, F. Enabling early detection of lithium-ion battery degradation by linking electrochemical properties to equivalent circuit model parameters. *J. Energy Storage* **2022**, *50*, 104213. [CrossRef]
- 18. Rahman, M.A.; Anwar, S.; Izadian, A. Electrochemical model parameter identification of a lithium-ion battery using particle swarm optimization method. *J. Power Sources* **2016**, *307*, 86–97. [CrossRef]
- Chan, H.L.; Sutanto, D. A new battery model for use with battery energy storage systems and electric vehicles power systems. In Proceedings of the 2000 IEEE Power Engineering Society Winter Meeting. Conference Proceedings (Cat. No.00CH37077), Singapore, 23–27 January 2000; Institute of Electrical and Electronics Engineers Inc.: New York, NY, USA, 2000; pp. 470–475. [CrossRef]
- 20. He, H.; Xiong, R.; Guo, H.; Li, S. Comparison study on the battery models used for the energy management of batteries in electric vehicles. In *Energy Conversion and Management*; Elsevier Ltd.: Amsterdam, The Netherlands, 2012; pp. 113–121. [CrossRef]

- 21. Campagna, N.; Castiglia, V.; Miceli, R.; Mastromauro, R.A.; Spataro, C.; Trapanese, M.; Viola, F. Battery models for battery powered applications: A comparative study. *Energies* **2020**, *13*, 85. [CrossRef]
- 22. Berger, M.; Kocar, I.; Farantatos, E.; Haddadi, A. Modeling of Li-ion battery energy storage systems (BESSs) for grid fault analysis. *Electr. Power Syst. Res.* 2021, 196, 107160. [CrossRef]
- 23. Pichetjamroen, A.; Chindamanee, P.; Teerakawanich, N.; Somakettarin, N. A Study on Performances of Flexible Power Control with Empirical Lithium-ion Battery Modeling in PV Power Systems. In Proceedings of the 19th International Conference on Electrical Engineering/Electronics, Computer, Telecommunications and Information Technology, ECTI-CON 2022, Prachuap Khiri Khan, Thailand, 24–27 May 2022; Institute of Electrical and Electronics Engineers Inc.: New York, NY, USA, 2022. [CrossRef]
- 24. Perinov; Garniwa, I.; Hudaya, C.; Sudiarto, B. Comparison of Battery Models for Battery Energy Storage System Development. J. Phys. Conf. Ser. 2021, 1858, 012046. [CrossRef]
- 25. Hussein, A.A.-H.; Batarseh, I. An overview of generic battery models. In 2011 IEEE Power and Energy Society General Meeting; IEEE: New York, NY, USA, 2011; pp. 1–6.
- Nebuloni, R.; Meraldi, L.; Bovo, C.; Ilea, V.; Berizzi, A.; Sinha, S.; Tamirisakandala, R.B.; Raboni, P. A hierarchical two-level MILP optimization model for the management of grid-connected BESS considering accurate physical model. *Appl. Energy* 2023, 334, 120697. [CrossRef]
- 27. Shamarova, N.; Suslov, K.; Ilyushin, P.; Shushpanov, I. Review of Battery Energy Storage Systems Modeling in Microgrids with Renewables Considering Battery Degradation. *Energies* **2022**, *15*, 6967. [CrossRef]
- Weniger, J.; Tjaden, T.; Quaschning, V. Sizing of residential PV battery systems. In *Energy Procedia*; Elsevier Ltd.: Amsterdam, The Netherlands, 2014; pp. 78–87. [CrossRef]
- Marchi, B.; Pasetti, M.; Zanoni, S. Life Cycle Cost Analysis for BESS Optimal Sizing. In *Energy Procedia*; Elsevier Ltd.: Amsterdam, The Netherlands, 2017; pp. 127–134. [CrossRef]
- Etherden, N.; Bollen, M.H.J. Dimensioning of energy storage for increased integration of wind power. *IEEE Trans. Sustain. Energy* 2013, 4, 546–553. [CrossRef]
- Ravishankar, R.; Vrudhula, S.; Rakhmatov, D. Battery Modeling for Energy-Aware System Design. Computer 2003, 36, 77–87. [CrossRef]
- 32. Wu, D.; Ma, X. Modeling and Optimization Methods for Controlling and Sizing Grid-Connected Energy Storage: A Review. *Curr. Sustain./Renew. Energy Rep.* 2021, *8*, 123–130. [CrossRef]
- 33. Wu, D.; Balducci, P.; Crrawford, A.; Mongrid, K.; Ma, X. Building Battery Energy Storage System Performance Data into economic assessment. In *IEEE Power & Energy Society General Meeting (PESGM)*; IEEE: New York, NY, USA, 2020; pp. 1–5.
- 34. Nguyen, T.A.; Copp, D.A.; Byrne, R.H.; Chalamala, B.R. Market Evaluation of Energy Storage Systems Incorporating Technology-Specific Nonlinear Models. *IEEE Trans. Power Syst.* 2019, 34, 3706–3715. [CrossRef]
- Hannan, M.A.; Faisal, M.; Ker, P.J.; Begum, R.A.; Dong, Z.Y.; Zhang, C. Review of optimal methods and algorithms for sizing energy storage systems to achieve decarbonization in microgrid applications. In *Renewable and Sustainable Energy Reviews*; Elsevier Ltd.: Amsterdam, The Netherlands, 2020; Volume 131. [CrossRef]
- Fernández-Blanco, R.; Dvorkin, Y.; Xu, B.; Wang, Y.; Kirschen, D.S. Optimal Energy Storage Siting and Sizing: A WECC Case Study. IEEE Trans. Sustain. Energy 2017, 8, 733–743. [CrossRef]
- 37. Shuai, H.; Fang, J.; Ai, X.; Tang, Y.; Wen, J.; He, H. Stochastic optimization of economic dispatch for microgrid based on approximate dynamic programming. *IEEE Trans. Smart Grid* **2019**, *10*, 2440–2452. [CrossRef]
- 38. Pozo, D. Linear battery models for power systems analysis. Electric. Power Syst. Res. 2022, 212, 108565. [CrossRef]
- Akram, U.; Khalid, M.; Shafiq, S. Optimal sizing of a wind/solar/battery hybrid grid-connected microgrid system. *IET Renew.* Power Gener. 2018, 12, 72–80. [CrossRef]
- Javadi, M.; Gong, Y.; Chung, C.Y. Frequency Stability Constrained BESS Sizing Model for Microgrids. *IEEE Trans. Power Syst.* 2023, 1–13. [CrossRef]
- Duman, A.C.; Erden, H.S.; Gönül, Ö.; Güler, Ö. Optimal sizing of PV-BESS units for home energy management system-equipped households considering day-ahead load scheduling for demand response and self-consumption. *Energy Build* 2022, 267, 112164. [CrossRef]
- 42. Powell, W. Approximate Dynamic Programming: Solving the Curses of Dimensionality Informs Computing Society Tutorial. 2008. Available online: http://www.castlelab.princeton.edu (accessed on 30 May 2023).
- 43. Wang, X.; Li, F.; Zhang, Q.; Shi, Q.; Wang, J. Profit-Oriented BESS Siting and Sizing in Deregulated Distribution Systems. *IEEE Trans. Smart Grid* 2023, 14, 1528–1540. [CrossRef]
- 44. Hannan, M.A.; Wali, S.B.; Ker, P.J.; Abd Rahman, M.S.; Mansor, M.; Ramachandaramurthy, V.K.; Muttaqi, K.M.; Mahlia, T.M.I.; Dong, Z.Y. Battery energy-storage system: A review of technologies, optimization objectives, constraints, approaches, and outstanding issues. *J. Energy Storage* **2021**, *42*, 103023. [CrossRef]
- Lampinen, J. Multiobjective Nonlinear Pareto-Optimization Multiobjective Nonlinear Pareto-Optimization A Pre-Investigation Report LAPPEENRANTA 2000. Available online: https://www.researchgate.net/publication/239560754 (accessed on 12 January 2023).
- 46. Truong, C.N.; Schimpe, M.; Bürger, U.; Hesse, H.C.; Jossen, A. Multi-use of stationary battery storage systems with blockchain based markets. In *Energy Procedia*; Elsevier Ltd.: Amsterdam, The Netherlands, 2018; pp. 3–16. [CrossRef]

- 47. Hjalmarsson, J.; Thomas, K.; Boström, C. Service stacking using energy storage systems for grid applications—A review. *J. Energy* Storage 2023, 60, 106639. [CrossRef]
- Yang, Y.; Bremner, S.; Menictas, C.; Kay, M. Battery energy storage system size determination in renewable energy systems: A review. In *Renewable and Sustainable Energy Reviews*; Elsevier Ltd.: Amsterdam, The Netherlands, 2018; Volume 91, pp. 109–125. [CrossRef]
- 49. Rancilio, G.; Bovera, F.; Merlo, M. Revenue Stacking for BESS: Fast Frequency Regulation and Balancing Market Participation in Italy. *Int. Trans. Electr. Energy Syst.* **2022**, 1894003. [CrossRef]
- 50. Brogan, P.V.; Best, R.; Morrow, J.; Duncan, R.; Kubik, M. Stacking battery energy storage revenues with enhanced service provision. *IET Smart Grid* **2020**, *3*, 520–529. [CrossRef]
- Alharbi, A.M.; Alsaidan, I.; Gao, W. Optimal Scheduling of Battery Energy Storage System Performing Stacked Services. In Proceedings of the IEEE Green Technologies Conference, Houston, TX, USA, 30 March–1 April 2022; IEEE: New York, NY, USA, 2022; pp. 110–115. [CrossRef]
- Zhang, Y.; Anvari-Moghaddam, A.; Peyghami, S.; Dragicevic, T.; Li, Y.; Blaabjerg, F. Optimal sizing of behind-the-meter BESS for providing stackable services. In Proceedings of the 2022 IEEE 13th International Symposium on Power Electronics for Distributed Generation Systems (PEDG 2022), Kiel, Germany, 26–29 June 2022; IEEE: New York, NY, USA, 2022. [CrossRef]
- Watson, D.; Hastie, C.; Gaudette, B.; Rodgers, M. Demonstrating Stacked Services of a Battery in a Wind R&D Park. *IEEE Trans.* Power Syst. 2018, 33, 1411–1419. [CrossRef]
- 54. Hameed, Z.; Træholt, C.; Hashemi, S. Investigating the participation of battery energy storage systems in the Nordic ancillary services markets from a business perspective. *J. Energy Storage* 2023, *58*, 106464. [CrossRef]
- 55. Tsagkou, A.S.; Kerasidis, E.D.; Doukas, D.I.; Labridis, D.P.; Marinopoulos, A.G.; Tengnér, T. Stacking Grid Services with Energy Storage Techno-Economic Analysis. In Proceedings of the IEEE Manchester PowerTech, Manchester, UK, 18–22 June 2017; pp. 1–6.
- 56. Liu, Y.; Du, W.; Xiao, L.; Wang, H.; Bu, S.; Cao, J. Sizing a Hybrid Energy Storage System for Maintaining Power Balance of an Isolated System with High Penetration of Wind Generation. *IEEE Trans. Power Syst.* **2016**, *31*, 3267–3275. [CrossRef]
- 57. Pelletier, S.; Jabali, O.; Laporte, G.; Veneroni, M. Battery degradation and behaviour for electric vehicles: Review and numerical analyses of several models. *Transp. Res. Part B Methodol.* **2017**, *103*, 158–187. [CrossRef]
- 58. Moncecchi, M.; Brivio, C.; Mandelli, S.; Merlo, M. Battery energy storage systems in microgrids: Modeling and design criteria. *Energies* **2020**, *13*, 2006. [CrossRef]
- 59. Ecker, M.; Nieto, N.; Käbitz, S.; Schmalstieg, J.; Blanke, H.; Warnecke, A.; Sauer, D.U. Calendar and cycle life study of Li(NiMnCo)O₂-based 18650 lithium-ion batteries. *J. Power Sources* **2014**, *248*, 839–851. [CrossRef]
- 60. Ramasamy, V.; Zuboy, J.; O'Shaughnessy, E.; Feldman, D.; Desai, J.; Woodhouse, M.; Basore, P.; Margolis, R.U.S. Solar Photovoltaic System and Energy Storage Cost Benchmarks, With Minimum Sustainable Price Analysis: Q1 2022. 2022. Available online: https://www.osti.gov/biblio/1891204 (accessed on 15 June 2023).
- Cole, W.; Frazier, A.W.; Augustine, C. Cost Projections for Utility-Scale Battery Storage: 2021 Update. 2021. Available online: https://www.nrel.gov/docs/fy21osti/79236.pdf (accessed on 15 June 2023).
- Rancilio, G.; Merlo, M.; Lucas, A.; Kotsakis, E.; Delfanti, M. BESS modelling: Investigating the role of auxiliary system consumption in efficiency derating. In Proceedings of the International Symposium on Power Electronics, Electrical Drives, Automation and Motion, Sorrento, Italy, 24–26 June 2020; pp. 1–6.
- Terna Mercato della Capacità—Disposizioni Tecniche di Funzionamento n.2. 2021. Available online: https://www.terna.it/it/sistema-elettrico/pubblicazioni/news-operatori/dettaglio/CM-Disposizioni-tecniche-di-funzionamento-DTF (accessed on 10 March 2023).
- 64. Gulotta, F.; Rossi, A.; Bovera, F.; Falabretti, D.; Galliani, A.; Merlo, M.; Rancilio, G. Opening of the Italian Ancillary Service Market to Distributed Energy Resources: Preliminary Results of UVAM project. In Proceedings of the HONET 2020—IEEE 17th International Conference on Smart Communities: Improving Quality of Life Using ICT, IoT and AI, Charlotte, NC, USA, 14–16 December 2020; IEEE: New York, NY, USA, 2020; pp. 199–203. [CrossRef]
- Bovera, F.; Blaco, A.; Rancilio, G.; Delfanti, M. Assessing the Accuracy of Different Machine Learning Classification Algorithms in Forecasting Results of Italian Ancillary Services Market. In Proceedings of the 2019 16th International Conference on the European Energy Market (EEM), Ljubljana, Slovenia, 18–20 September 2019.
- 66. Arera. Deliberazione 3 Giugno 2020 200/2020/R/EEL Approvazione del Regolamento, Predisposto da Terna S.P.A., Ai Sensi della Deliberazione Dell'autorità 300/2017/R/EEL, Relativo al Progetto Pilota per L'erogazione del Servizio di Regolazione Ultra-Rapida di Frequenza. 2020. Available online: https://www.arera.it/it/docs/20/200-20.htm (accessed on 25 February 2023).
- 67. Terna Mercato Della Capacità Rendiconto Degli Esiti—Asta Madre 2024. 2022. Available online: https://www.terna.it/it/sistema-elettrico/pubblicazioni/news-operatori/dettaglio/rendiconto-asta-capacity-market-2024 (accessed on 14 March 2023).
- 68. Terna Progetto Pilota Fast Reserve—Esiti Asta. 2020. Available online: https://www.terna.it/it/sistema-elettrico/pubblicazioni/ news-operatori/dettaglio/esiti-asta-Fast-reserve (accessed on 28 January 2023).
- 69. RSE Atlante Eolico RSE. Available online: https://atlanteeolico.rse-web.it/ (accessed on 7 June 2023).
- ARPA Puglia Temperature Puglia. Available online: https://www.arpa.puglia.it/pagina2839_meteo.html (accessed on 7 June 2023).

- 71. Gestore Mercati Energetici (GME) Esiti Mercato Elettrico. Available online: https://www.mercatoelettrico.org/it/ (accessed on 7 June 2023).
- 72. Afman, M.; Hers, S.; Scholten, T. Energy and Electricity Price Scenarios 2020-2023-2030. 2017. Available online: chrome-extension: //efaidnbmnnibpcajpcglclefindmkaj/https://cedelft.eu/wp-content/uploads/sites/2/2021/04/CE_Delft_3H58_Energy_and_electricity_price_scenarios_DEF.pdf (accessed on 7 June 2023).
- Mercier, T.; Olivier, M.; De Jaeger, E. The value of electricity storage arbitrage on day-ahead markets across Europe. *Energy Econ.* 2023, 123, 106721. [CrossRef]
- 74. Lazard LCOE Lazard. 2023. Available online: https://www.lazard.com/research-insights/2023-levelized-cost-of-energyplus/ (accessed on 7 June 2023).

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.